Author's response to reviews

Title: Comparison of two dengue NS1 rapid tests for sensitivity, specificity and relationship to viraemia and antibody responses

Authors:

Vianney Tricou (vtricou@oucru.org)
Hang TT Vu (hangytt@oucru.org)
Nhu VN Quynh (nhuyovo2611@yahoo.com.vn)
Chau VV Nguyen (chaunnv@oucru.org)
Hien T Tran (hientt@oucru.org)
Jeremy Farrar (jfarrar@oucru.org)
Bridget Wills (bwills@oucru.org)
Cameron P Simmons (csimmons@oucru.org)

Version: 4  Date: 19 May 2010

Author's response to reviews:

Dear Editors,

I have done all the requested formatting changes (major and minor révisions).

In addition I have done a few corrections:
- BioRad became Bio-Rad.
- Legend of figure 2 has been corrected (a details was missing).
- Hematological became haematological.

Warmest regards,

Vianney Tricou.